Author's response to reviews

Title: Effective cardiac resynchronization therapy for an adolescent patient with dilated cardiomyopathy seven years after mitral valve replacement and septal anterior ventricular exclusion

Authors:

Takahiro Mima (takamima@kuhp.kyoto-u.ac.jp)
Shiro Baba (shibaba@kuhp.kyoto-u.ac.jp)
Noritaka Yokoo (nyokoo@kuhp.kyoto-u.ac.jp)
Shinji Kaichi (skaichi@kuhp.kyoto-u.ac.jp)
Takahiro Doi (t_doiji@kuhp.kyoto-u.ac.jp)
Hiraku Doi (hiraku@kuhp.kyoto-u.ac.jp)
Toshio Heike (heike@kuhp.kyoto-u.ac.jp)

Version: 3 Date: 29 May 2010

Author's response to reviews:

Vipin Zamvar and David Taggart
Editors-in-Chief
Journal of Cardiothoracic Surgery

Manuscript Number: 4558772123679737 (Revised Version)

Title: Effective cardiac resynchronization therapy for an adolescent patient with dilated cardiomyopathy seven years after mitral valve replacement and septal anterior ventricular exclusion

Authors: Takahiro Mima, Shiro Baba, Noritaka Yokoo, Shinji Kaichi, Takahiro Doi, Hiraku Doi, and Toshio Heike

Dear Editors,

Thank you very much for your acceptance of publishing our manuscript in Journal of Cardiothoracic Surgery. We have made formatting changes as you request.

Major revisions (we require the author to make these changes)

Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend.

We have provided figure titles as follows.
Figure 1. ECG before and after the CRT placement.
Figure 2. The chest X-ray (PA and Lateral projection) after the CRT placement.
Figure 3. Echocardiogram before and after the CRT placement.
Figures: Figure 3 is cited in the text before figure 2. Please reorder the figure citation so that they are cited in consecutive order.

We have reordered the figure citation. We have changed Figure 2 with Figure 3.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title page: please change E-mail lists to E-mail addresses, and use initials instead of full names

We have changed E-mail lists to E-mail addresses, and used initials instead of full names.

Affiliations: please remove the following details from your affiliations: `54 Kawahara-cho, Shogoin, Sakyo-ku, Kyoto 606-8507, Japan`.

We have removed the details from our affiliations: 54 Kawahara-cho, Shogoin, Sakyo-ku, Kyoto 606-8507, Japan

Highlighting/tracking: Please remove all highlighting and tracking/coloured text from the manuscript.

We have removed all highlighting and tracking/coloured text from the manuscript.

Thank you

Sincerely,

Shiro Baba, M.D., Ph.D.
Department of Pediatrics, Graduate School of Medicine, Kyoto University, 54 Kawara-cho, Shogoin, Sakyo-ku, Kyoto 606-8507, Japan
Tel: +81-75-751-3290, Fax: +81-75-752-2361,
E-mail: shibaba@kuhp.Kyoto-u.ac.jp